Author's response to reviews

Title: Association Analysis of Chromosome 1 Migraine Candidate Genes

Authors:

Francesca Fernandez (f.fernandez@griffith.edu.au)
Robert P Curtain (robcurtain@hotmail.com)
Natalie J Colson (n.colson@griffith.edu.au)
Micky Ovcaric (m.ovcaric@griffith.edu.au)
John MacMillan (john_macmillan@health.qld.gov.au)
Lyn R Griffiths (l.griffiths@griffith.edu.au)

Version: 5 Date: 23 August 2007

Author's response to reviews: see over
1) Text modifications:

- **Title section**: All the unnecessary capital letters have been removed

- **Author list**: The list of authors in the manuscript has been changed at the right format for submission e.g. forename, initial (if any), surname.

- **Affiliations**: The italic font has been changed by the regular font.

- **Key Words** section has been removed.

- **Subjects** section has been replaced for a sub-heading section.

The “Competing Interests” has been replaced by “Competing interests”

The Author's contributions has been replaced by “Authors' contributions”

- **References**: The references have been organized by ascending numerical order of citing in the text.

2) Table modifications:

- **Table titles**: The full stops (from the end of the table title) have been removed for all the tables.

- **Tables**:

The Tables 3, 4 and 5 have been divided into cells, as suggested.

In Table 2, the hard returns in the last column have been removed. The legend following “Positioning of genes (cM) based on NCBI Map Viewer using 'deCODE' and 'Genes On Sequence mapping data” has been moved below the table.

The Table 3 has been organized in one unique table with 4 main parts. The headings ATP 1A4 SNP (rs 6427504), CAS Q1 SNP (rs 3747623), KCN J9 SNP (rs 2494211), KCN J10 SNP (rs 1130183) have been moved inside the table body by placing each of them in a row immediately above each of the 'Migraine'.

The text in the brackets ‘MO, migraine without aura; MA, migraine with aura” has been moved to below the table as a table 3 legend.

In Table 4, all the unnecessary capitals have been removed in the title and in the last column of the table.